Author’s response to reviews

Title: What works for whom in the management of diabetes in people living with dementia: a realist review

Authors:

Frances Bunn (f.bunn@herts.ac.uk)
Claire Goodman (c.goodman@herts.ac.uk)
Peter Reece-Jones (p.jones@bangor.ac.uk)
Bridget Russell (b.s.russell@herts.ac.uk)
Daksha Trivedi (d.trivedi@herts.ac.uk)
Alan Sinclair (ajsinclair@diabetesfrail.org)
Antony Bayer (Bayer@cardiff.ac.uk)
Greta Rait (g.rait@ucl.ac.uk)
Jo Rycroft-Malone (j.rycroft-malone@bangor.ac.uk)
Christopher Burton (c.burton@bangor.ac.uk)

Version: 2 Date: 03 Jul 2017

Author’s response to reviews:

Dear Editor

Thank you for your email.

A competing interest statement has been added to the manuscript.

As agreed the existing RAMASES checklist is sufficient.

Best wishes

Frances